Author's response to reviews

Title: Audit and Feedback and Clinical Practice Guideline Adherence: Making Feedback Actionable

Authors:
Sylvia J. Hysong (sylvia.hysong@med.va.gov)
Richard G. Best (rbest@satx.rr.com)
Jacqueline A. Pugh (pugh@uthscsa.edu)

Version: 4 Date: 27 April 2006

Author's response to reviews:

This is the final copy-edited version for publication, with all the formatting changes requested on 4/27 completed. Requested changes detailed below:

1. Title - Please remove all unnecessary capitalisation from the title, both in the manuscript file and the submission system. SJH: Only the first word and the word after the colon are now capitalized.

2. Affiliations - Each affiliation should be written in full, even if it is more than one department within the same Institution, and be provided in the following format: department, institute, city, country. SJH: Author affiliations are now listed in the manuscript as follows:
   - Houston Center for Quality of Care and Utilization Studies, Michael E. DeBakey VA Medical Center, Houston, Texas, USA
   - Department of Medicine - Health Services Research Section, Baylor College of Medicine, Houston, Texas, USA
   - Healthcare Solutions Division, Lockheed Martin Information Technology, San Antonio, Texas, USA
   - Veterans Evidence-Based Research Dissemination and Implementation Center, South Texas Veterans Health Care System, San Antonio, Texas, USA
   - Department of Medicine, University of Texas Health Science Center at San Antonio, San Antonio, Texas, USA

3. Authors' contributions - For the Authors' contributions we suggest the following kind of format (please use initials to refer to each author's contribution): AB carried out the molecular genetic studies, participated in the sequence alignment and drafted the manuscript. JY carried out the immunoassays. MT participated in the sequence alignment. ES participated in the design of the study and performed the statistical analysis. FG conceived of the study, and participated in its design and coordination. All authors read and approved the final manuscript. SJH: Author contribution has been edited to follow the suggested format.

4. References - The reference list should contain all authors' names, regardless of how many there are. The term "et al" should not be used. SJH: References now contain full author lists.

5. Tables - Unfortunately, we will be unable to display shading or colour within the tables. Can we therefore ask you to use another method of highlighting the shaded text e.g. by making it bold. SJH: Shading has been removed from tables in the appendix, and has been replaced by bold text of a different font.

6. Figures - It is important for the final layout of the manuscript that the figures are cropped as closely as possible to minimise white space around the image. SJH: Figure 1 has been cropped per your guidelines at http://www.biomedcentral.com/info/ifora/cropping.